

Geophysical Research Letters



RESEARCH LETTER

10.1029/2020GL089627

Key Points:

- Lithosphere rigidity has a control on the geometry of large rivers near passive margins
- Flexure is responsible for the formation of axial rivers running parallel to margins
- This explains why continental interiors drain through narrow, widely spaced outlets into oceans

Supporting Information:

Figure S1

Correspondence to:

R. Jiao, rjiao@uvic.ca

Citation:

Jiao, R., Braun, J., & Kravitz, K. (2020). Isostatic control of axial rivers and large drainage basins on passive margins. *Geophysical Research Letters*, 47, e2020GL089627. https://doi.org/ 10.1029/2020GL089627

Received 1 JUL 2020 Accepted 2 AUG 2020 Accepted article online 10 AUG 2020

 $\textcircled{\sc 0}2020.$ The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Isostatic Control of Axial Rivers and Large Drainage Basins on Passive Margins

Ruohong Jiao^{1,2}, Jean Braun¹, and Katherine Kravitz^{1,3}

¹Helmholtz Centre Potsdam, GFZ German Research Centre for Geosciences, Potsdam, Germany, ²School of Earth and Ocean Sciences, University of Victoria, Victoria, British Columbia, Canada, ³Now at Department of Earth and Atmospheric Sciences, Indiana University, Bloomington, IN, USA

Abstract More than half of the world's large rivers flow towards the ocean crossing passive continental margins. Here using an analytical solution and numerical models, we demonstrate that on passive margins, river basins may be integrated by major margin-parallel channels, which form as a flexural isostatic response of the lithosphere to mechanical/erosional unloading along the margin. We analyzed the downstream courses of large rivers flowing across the passive margins and find that the majority of them (31 of 36) have major margin-parallel channels. Occurrences of these channels are generally consistent with the model predictions, although the exact locations and geometry of these rivers may also be controlled/changed by other factors. Our results suggest that the lithosphere strength has an important control on the geometry of large river systems on passive margins, linking the evolution and routing of the Earth's freshwater systems to its deep interior dynamics.

Plain Language Summary Tectonically stable continents are often drained by large rivers, which flow to the ocean through a relatively narrow main stem. We investigated why these drainage patterns develop and specifically, why these rivers do not flow straightly to their outlets. Along passive margins, we find that most of the world's biggest rivers develop channels parallel to the margin before they enter the ocean and the occurrences of these channels are responsible for the formation of large drainage basins. The locations of these channels indicate that they were created and maintained by isostatic uplift in response to the erosional removal of the surface rocks. This uplift along the margin creates a depression further inland that causes the rivers to flow parallel to the margin and increases the size of the river basin. Our proposed mechanism demonstrates how the structure and dynamics of the Earth's interior could impact the cycles in the hydrosphere and thus biosphere at the surface.

1. Introduction

The geometry of river networks determines how water runoff is routed across continents and therefore has a direct influence on freshwater input into the ocean, carbon cycling, and the pattern of freshwater ecosystems. In steep mountainous areas, stream orientation is mostly controlled by local-scale topographic relief and active structures (e.g., Clark et al., 2004). However, many continents are dominated by low-relief interiors, where river networks often do not follow the topographic gradient (Black et al., 2017). It has been argued that low-amplitude topography caused by deep mantle processes may control the geometry of some drainage systems at the Earth's surface (e.g., Cox, 1989), but no universal pattern has been observed to support this hypothesis.

In low-relief cratonic areas, rivers connect to the ocean across passive margins. Drainage networks along passive margins are typically characterized by a dichotomy of river basin sizes, with small coastal rivers that drain directly to the ocean separated from large river basins that drain the continental interior surrounded by a rim of high topography. This topographic bulge is thought to form during continental breakup as a flexural isostatic response to continental stretching (Braun & Beaumont, 1989) and/or progressive erosion of an escarpment or topographic step (Gilchrist & Summerfield, 1990) that separates the coastal plain from the relatively flat, elevated continental interior (e.g., Figures 1a and 1b).

Along most passive margins, this escarpment or bulge is a drainage divide. Evolution of this drainage divide is determined by the competition between denudation and isostatic uplift (Braun, 2018; Tucker &





Figure 1. (a) Example of the axial rivers in the Orange River basin along the Atlantic passive margin of southern Africa. Pixels of the river channels are color coded according to the δ values, which are calculated as $\delta = |cos(\theta)|$, where θ is the angle between the downstream flow direction and the direction towards the margin. d_a is the distance to the margin, that is the shortest distance from the river pixel to the transitional crust. Green line marks the location of the transitional crust with the thickness of 20 km. Box indicates the location of the topographic profile in b. (b) Topography along a profile within a swath. Black line indicates the median topography. (c) Sketch illustrating the flexural responses to erosion along a passive margin and the theoretical position of the topographic depression where an axial river may form.

Slingerland, 1994). On the one hand, river incision and hillslope erosion reduce the elevation of the escarpment/bulge. On the other hand, the lowering of the surface elevation is partially compensated by flexural isostatic rebound triggered by erosional unloading (Figure 1c). These processes act to maintain the escarpment/bulge as a drainage divide and preserve it for tens of millions of years or longer after rifting, even if it has migrated several tens to hundreds of kilometers landward from its original position (Gilchrist & Summerfield, 1990).

Flexural isostasy which works to create and maintain the escarpment also results in low-amplitude subsidence of the continental area behind it (Figure 1c) (Sacek et al., 2012). We postulate that this topographic low leads to the formation of a valley that parallels the passive margin and truncates the inland river network, diverting flow to a large, margin-parallel river. This "axial river" collects water and sediment from the truncated tributaries that drain expansive portions of the continental interior. Unable to breach the topographic barrier of the escarpment, it flows parallel to the coast until it reaches a coastal river that has incised through the topographic bulge to capture its drainage (Figure 2). By combining multiple river flows into a large one, the occurrence of an axial river significantly increases the size of the continental interior drainage basins. Therefore, we propose that this flexure-controlled continent-scale drainage network reorganization is responsible for the first-order geometry of major river basins on large, tectonically stable continents, such that the geometry of most rivers draining continental interiors can be predicted from the strength of the underlying lithosphere which ultimately determines the scale and width of its flexural isostatic response.

In this paper, we test this hypothesis by comparing the locations of the axial rivers (1) predicted by an analytical solution of the flexure of the lithosphere (Section 2), (2) simulated by numerical landscape evolution models (Section 3), and (3) observed in the largest river basins on Earth (Section 4). The results support our hypothesis and indicate that the strength of the lithosphere is a fundamental component to the drainage network geometry on passive margins, although the detailed patterns are also controlled by other parameters.

2. Topographic Depression Due to Lithospheric Flexure

On a continental margin, erosion of a narrow column of the lithosphere can be considered as a line load:

$$V_0 = \rho_l g h_e \Delta x,\tag{1}$$





Figure 2. Topography produced by a landscape evolution model at three time steps, showing the development of an axial river after continental rifting. Models were run on 241×241 grids with 5-km spacing. The effective elastic thickness (T_e) of the model lithosphere is 30 km.

in which ρ_l is the lithospheric density, g is the gravitational acceleration, h_e is the eroded thickness, and Δx is the width of the column. At the incipient stage of continental rifting, we assume that the erosion of the margin was concentrated at a position x = 0, where the initial escarpment developed. Assuming that the continental lithosphere acts as thin elastic plate of effective elastic plate thickness T_e , vertical surface displacement of the lithosphere caused by erosion is given by (Turcotte & Schubert, 1982)

$$u(x) = \frac{V_0 \alpha^3}{8D} \exp\left(-\frac{x}{\alpha}\right) \left(\cos\frac{x}{\alpha} + \sin\frac{x}{\alpha}\right),\tag{2}$$

in which α is the flexural wavelength given by

$$a = \sqrt[4]{\frac{4D}{\rho_a g}},\tag{3}$$

where ρ_a is asthenospheric rock density and D the flexural rigidity of the lithosphere defined by

0

$$D = \frac{ET_e^3}{12(1-\nu^2)},$$
(4)

where E and v are elastic parameters, namely Young's modulus and Poisson's ratio.

As we consider the load is concentrated at x = 0, which makes $\Delta x \to 0$, it is reasonable to assume that $\Delta x \ll \alpha$. In this case, the solution expressed in Equation 2 has the shape as shown in Figure 1c. It not only explains the uplift of the coastal region and the formation of the bulge/escarpment observed along many passive margins but also predicts the formation of a depression at a distance $\pi \alpha$ of the region of maximum erosion. As suggested by Sacek et al. (2012), a small amplitude, secondary bulge may also form, which will further contribute to the formation of an axial river on the landward side of and running parallel to the escarpment.

Assuming that the initial escarpment was developed at a tectonic hinge line that parallels the margin, this line should mark a significant transition in the crustal thickness and topography (e.g., Karner & Driscoll, 1999). The position of the axial river should be located at a distance

$$d_a = \pi \alpha \tag{5}$$

from the crustal transition line. This simple model predicts a power-law relationship between the locations of axial rivers along passive margins and the effective elastic thickness of the underlying lithosphere:

$$d_a \propto T_e^{4/3}.$$
 (6)

Therefore, in the case where isostatic adjustment from sediment loading can be neglected, the position of axial rivers is solely dependent on the flexural strength of the lithosphere, that is its distance to margin, d_a , is proportional to the effective elastic thickness raised to a power, $T_e^{4/3}$. Note that for simplification in the following text, we term the location of the hinge line of transitional crust as the margin of the continental plate.





Figure 3. Map views of landscape evolution models showing drainage networks on passive margins at 50 Ma after rifting, assuming T_e values of 30, 100 km, and ∞ (no isostasy), respectively. The models show that following the continental rifting, isostatic uplift can create axial rivers and large drainage basins, and the locations of the axial rivers are dependent on the lithosphere rigidity.

3. Landscape Evolution Modeling

The above prediction is dependent on important assumptions that the erosional unloading at the initial escarpment is significantly more than anywhere else on the margin and sediment deposition in the fluvial systems can be neglected. To test these assumptions and improve the predictions, we use a landscape evolution model to predict how the landscape responds to the distributed erosion and sedimentation along a passive margin following continental rifting and breakup. We use the approach developed by Braun and Willett (2013) and Yuan et al. (2019) to solve the basic equations governing the rate of fluvial incision, sediment deposition, and hillslope processes (Ahnert, 1977; Davy & Lague, 2009; Whipple & Tucker, 1999):

$$\frac{\partial h}{\partial t} = U - K_f A^m S + \frac{G}{A} \int_A \left(U - \frac{\partial h}{\partial t} \right) dA + K_d \nabla^2 h, \tag{7}$$

where *h* is topographic elevation, *t* is time, *A* is upstream drainage area, *S* is local topographic slope, *G* is a dimensionless number controlling the balance between erosion and sediment deposition in rivers (see Yuan et al., 2019, for details), $\nabla^2 h$ is surface curvature, and K_f and K_d are fluvial erosion and diffusion coefficients, respectively. The values of these rate coefficients are variable as they are likely dependent on lithology, rock erodibility, and/or climate. In this equation, *U* is the uplift rate due to erosional unloading, which we obtain by solving the biharmonic equation governing the flexure of a thin elastic plate (Turcotte & Schubert, 1982):

$$D\frac{\partial^4 U}{\partial x^4} + D\frac{\partial^4 U}{\partial y^4} + 2D\frac{\partial^4 U}{\partial x^2 \partial y^2} = (\rho_l - \rho_a)gU + \rho_l g\frac{\partial h}{\partial t}.$$
(8)

We solved these coupled equations in a 1,200 × 1,200 km² domain at 5-km resolution and started each model with a 500- to 700-m-high flat plateau to which we added an infinitesimal tilt to force the development of an initial drainage network flowing towards the left-hand side of the model representing a rifted/passive margin. We ran the models for 50 Myr using 10-kyr long time steps and different randomly selected values of K_f (10⁻⁷ – 10⁻⁵ m^{0.2}/yr), K_d (0–0.1 m²/yr), and T_e (20–120 km). We tested landscapes dominated by detachment-limited rivers and that with sediment deposition by prescribing *G* in Equation 7 at 0 and 1, respectively; *m* was fixed at 0.4. For the flexure model (Equations 4 and 8), *E*, v, ρ_l , and ρ_a were prescribed at 10¹¹ Pa, 0.25, 2,800 km/m³ and 3,200 kg/m³, respectively. We performed >50,000 model experiments for the settings without (*G* = 0) and with (*G* = 1) sediment deposition, respectively. For each model, the shape of river networks is recorded at every 10-Myr interval (e.g., Figure 2).

In order to determine the position of the axial channels, we estimated the river flow direction relative to the orientation of the margin. For each pixel of the river channel, we introduce a measure,

$$\delta = |\cos(\theta)|,\tag{9}$$

where θ is the angle between the observed downstream direction and the direction normal to the margin (Figure 1a); the former is calculated over a critical downstream length of 100 km. A δ value of 1 indicates that the river channel is perpendicular to the margin, whereas a value of 0 corresponds to channels flowing parallel to the margin. Assuming that the river network branching is purely topological and not controlled





Figure 4. Map showing the world's large rivers flowing towards passive (dark blue) and active (light blue) margins. Gray lines depict plate boundaries (Bird, 2003). Green lines depict locations of transitional crust (thickness of 20 km) with no significant tectonic deformation. Estimates of the effective elastic thickness (T_e) of the continental lithosphere are from Audet and Bürgmann (2011).

by local topography, the bifurcation angle between any two branches should be around 72° (Devauchelle et al., 2012), which should lead to the formation of branches with δ values around 0.8. We therefore used a low δ value (<0.3) to define the occurrence of axial rivers on the modeled landscape (e.g., Figure 3).

Our modeling results predict that axial rivers form on >90% of the modeled landscapes, which are located in a wide range of locations relative to the initial position of the escarpment. By summarizing the distances from the margin to the first occurrences of the axial rivers and plotting these distances versus the model T_e , we find that the highest densities of points representing the model outputs are distributed along the theoretical line predicted by the 1-D analytical solution (Figure 5), albeit that the model results with continental deposition are more dispersed.

4. Axial Channels of the Largest Rivers

In this section, we further test the control of lithosphere flexure on the formation of axial rivers by examining the downstream courses of the 56 rivers with the largest drainage areas that are located between 40°N and 40°S (Figure 4). High-latitude drainage systems are excluded from our analysis, as we wish to focus on basins dominated by fluvial processes. Of these 56 rivers 36 flow towards passive margins, where we consider that the perturbation of topography by tectonic activities is minimal. As described above, we use the δ value of a pixel on the river channel to determine its flow direction relative to the orientation of the margin. To perform the analysis, we used an upscaled (1/8° resolution) river network model (Wu et al., 2012) and calculated δ using critical downstream lengths of 50, 100, and 150 km. For most rivers, this choice of the critical segment length had little influence on our estimate of the location of low- δ segments (see Figure S1 for full details), so here we present only the results using a critical segment length of 100 km. We restricted our analysis to river segments with upstream drainage areas over 10,000 km² and within 1,000-km distance from the continental margin. To define the trace of the initial position of the escarpment, that is the transitional zone between continental and oceanic crusts, we used a simplification and considered it at the location where the crustal thickness is 20 ± 5 km (according to the model of Laske et al., 2013).

The results show that for all but one (Webi Jubba in Somalia) of the 36 rivers, high δ values occur near their outlets, indicating that these channels flow in a direction perpendicular to the margin. Upstream and further inland, rivers start to develop more dendritic patterns. The majority (31 out of 36) of the rivers we analyzed start to diverge or branch into channels with low δ values (<0.3), implying that the developments of their networks were not purely topological. To determine the positions of axial rivers on a margin, we binned the pixels of a river according to their distances to the margin and estimated a location where at least 40% channel pixels have low- δ values (<0.3). The threshold (40%) was determined by testing a range of thresholds to capture the average flow direction of the river channels, but for most rivers, using a threshold between 30% and 50% leads to no difference for determining the position of the axial river. The full results are shown in Figure S1, and summarized in Figure 5. To estimate the mechanical strength of the lithosphere, we calculated the mean T_e of the lithosphere within 500 km from the margin in a river's drainage area, using the results from spectral coherence analysis between topography and Bouguer gravity anomalies (Audet & Bürgmann, 2011).







Figure 5. Shortest distances from axial rivers to the continental margin versus lithosphere rigidity. (a) Data of rivers flowing towards the Atlantic margin of Africa (dot) and prediction of detachment-limited fluvial erosion models (shading). (b) Data of rivers on other passive margins (dot) and prediction of landscape evolution models considering sediment deposition (shading). Dashed lines indicate the theoretical distances from the margin to the flexural depression caused by a line erosional unloading at the margin. Effective elastic thickness (T_e) values are mean \pm standard deviation of the lithosphere in the downstream areas (<500 km to the margin) of the drainage basins (Audet & Bürgmann, 2011). The location of a margin is assumed at the position where the crust thickness is 20 ± 5 km; the crust thickness is according to Laske et al. (2013). Data points are color coded according to the sediment volume within a 100-km-wide swath from the river outlet to the margin, which is calculated using the model of Straume et al. (2019). Shading shows density contours of the individual model predictions from landscape evolution modeling.

Along the Atlantic margin of Africa, the determined distances between the margin and the axial rivers increase with the rigidity of the lithosphere (Figure 5a). The general consistency between theoretical prediction, modeling results, and the data of natural rivers in this region supports our hypothesis that flexural isostasy in response to erosion of the margin has a control on the drainage pattern. In contrast, along other passive margins such a relationship is not clearly expressed, and the data points show a considerable scattering around the theoretical prediction (Figure 5b). We suggest that the scattering is in part due to the non-deterministic nature of surface processes, but more importantly, the developments of axial rivers may be also controlled by other factors besides the isostatic response to the erosional unloading, such as active tectonics and sediment deposition offshore. To examine the potential link between the offshore sediment loading and the location of an axial river, we compiled the sediment volume in a 100-km-wide swath from the river outlet to the margin based on the model of Straume et al. (2019), which will be discussed in the following section.

5. Discussion and Implication

5.1. Effects of Sediment Loading and Other Factors

From the numerical modeling results, we observe that contours of the highest density of models are systematically above the theoretical curve, suggesting that locations of axial rivers may be further controlled by isostatic response to erosion over a region wider than the initial escarpment. This may reflect the erosion of the continent interior when the axial river started to form, or that in some cases, the continuous retreats of the escarpment and the locus of the highest erosion rate displaced the axial river landwards.

In addition to the flexural uplift induced by erosion of the escarpment, we consider sediment deposition to be another important process that affects the formation and location of the axial rivers. First, sediment deposition in the drainage basin, especially on the alluvial plains in the downstream part of the large rivers, complicates the distribution of mass unloading and thus the pattern isostatic rebound on the margin. Compared to the modeling results with detachment-limited rivers, models with onshore deposition predict a more dispersed locations of the axial rivers (Figure 5b), suggesting that redistribution of sediments contributes significantly to the variability of the data. Second, the accumulation of marine sediments on the margin has a flexural effect that is opposite in sign to the erosional unloading, and thus, the location of the resulting peripheral bulge associated with sediment loading can also be predicted using Equation 5. Therefore, the combined isostatic effect due to both the erosional unloading and the sediment loading on the margin will drive the locations of axial rivers further inland from the $\pi \alpha$ distance from the margin.

Compilation of the sediment accumulation on the shelf show that on a margin with high sediment volume (e.g., the Amazon; Figure 5b), the axial river usually occurs further inland.

The scattering of data from natural rivers is also contributed by the efficiency of the surface processes. We notice, for example, that due to the low magnitude of flexural depression, the drainage network reorganization in a substantial amount (<10%) of model simulations yield no axial rivers, which is consistent with our observations that some large rivers on passive margins (e.g., the Colorado River in Texas) do not have major axial channels. Furthermore, contrasting rock erodibility between different lithological units can also lead to the formation of axial rivers. For example, in the drainage basins of the Ogooue and Mahanadi rivers, the axial river channels developed in the relatively weak metasedimentary belts that are parallel to the margin, which are the West Congo belt (Thiéblemont et al., 2018) and Khondalite zone (Dobmeier & Raith, 2003), respectively.

The drainage patterns of some rivers may also be perturbed by regional tectonics, even though the rivers do not flow towards the convergent margins. A clear outlier in our analysis is the Rio Grande. We suspect that, despite having its outlet along a passive margin on the Gulf of Mexico, geometry of the river network is likely controlled by tectonic uplift as the majority of its upstream course lies in or near the southern Rocky Mountains. Along the east coast of Africa, development of the East African Rift lifted the topography and created grabens, which have influenced the drainage patterns of rivers in the area (e.g., the Zambezi and Limpopo).

5.2. Long-term Stability of the Axial Rivers

The modeling results also suggest that the impact of lithospheric flexure on river networks can be preserved tens of millions of years after rifting (Figure 2). Once an axial river develops, it captures the drainage areas of several upstream inland rivers, therefore aggregating their discharge/drainage. This results in high erosion rates and the entrenchment of its position in the landscape, a process amplified by the isostatic uplift of surrounding areas. Consequently, the position of the axial river is likely to remain at its initial position set during continental break-up even in situations where the initial topographic relief has been strongly degraded or valley growth has extended further into the plateau.

A clear example of the long-term stability of axial river channels is expressed in the present-day geometry of the Cuanza River. The river had lost a significant portion of its drainage area to the present-day neighboring Congo River during a period of drainage reorganization (Leturmy et al., 2003), which was potentially driven by sub-lithosphere mantle upwelling (Lithgow-Bertelloni & Silver, 1998) or late marginal flexure (Lucazeau et al., 2003). Yet, segments of its axial channels have been preserved at locations consistent with our predictions (Figure 5), perhaps set during the late Cretaceous following rifting along the Atlantic margin of Africa.

5.3. Potential Impacts

The formation of axial rivers causes continental-scale river drainage reorganization, which has led to the development of most of the largest river systems on the Earth. This, in turn, has had strong consequences on many aspects of the Earth's system. For example, the routing of continental freshwater export to the ocean affects the relative salinity of the large ocean basins, which controls planetary-scale ocean circulation (Maffre et al., 2018; Sinha et al., 2012). Without lithospheric flexure, the freshwater flux to the ocean would be more evenly distributed along continental coastlines, which would likely decrease the salinity gradient between ocean basins and lead to more sluggish global ocean circulation.

The geometry and evolution of continental drainage systems are also known to exert a strong control on the biodiversity and species richness within freshwater ecosystems. For example, it has been documented across many river systems that the number of freshwater species per unit area increases with the river discharge (McGarvey & Ward, 2008; Xenopoulos & Lodge, 2006). Connecting smaller rivers into a larger drainage basin can also potentially increase fish species richness. For example, higher fish species richness has been observed downstream of confluences (Fernandes et al., 2004; Osborne & Wiley, 1992), and tributaries are also known to support more species than same-sized rivers directly connected to the ocean (Oberdorff et al., 1997). The stability of large river networks is also conducive to higher freshwater fish species richness (Griffiths, 2018). Therefore, our findings that continental river network geometry is preserved long after rifting suggests that lithospheric flexure has facilitated species diversification.



Data Availability Statement

The global river network data (available at http://files.ntsg.umt.edu/data/DRT) and effective elastic thickness estimates were published by Wu et al. (2012) and Audet and Bürgmann (2011), respectively.

References

Ahnert, F. (1977). Some comments on the quantitative formulation of geomorphological processes in a theoretical model. *Earth Surface Processes*, *2*(2-3), 191–201. https://doi.org/10.1002/esp.3290020211

Audet, P., & Bürgmann, R. (2011). Dominant role of tectonic inheritance in supercontinent cycles. *Nature Geoscience*, 4(3), 184–187. https://doi.org/10.1038/ngeo1080

Bird, P. (2003). An updated digital model of plate boundaries. Geochemistry, Geophysics, Geosystems, 4(3), 1027. https://doi.org/10.1029/2001GC000252

Black, B. A., Perron, J. T., Hemingway, D., Bailey, E., Nimmo, F., & Zebker, H. (2017). Global drainage patterns and the origins of topographic relief on Earth, Mars, and Titan. *Science*, *356*(6339), 727–731. https://doi.org/10.1126/science.aag0171

Braun, J. (2018). A review of numerical modeling studies of passive margin escarpments leading to a new analytical expression for the rate of escarpment migration velocity. *Gondwana Research*, 53, 209–224. https://doi.org/10.1016/j.gr.2017.04.012

Braun, J., & Beaumont, C. (1989). A physical explanation of the relation between flank uplifts and the breakup unconformity at rifted continental margins. *Geology*, *17*(8), 760–764. https://doi.org/10.1130/0091-7613(1989)017<0760:APEOTR>2.3.CO;2

Braun, J., & Willett, S. D. (2013). A very efficient O(n), implicit and parallel method to solve the stream power equation governing fluvial incision and landscape evolution. *Geomorphology*, *180-181*, 170–179. https://doi.org/10.1016/j.geomorph.2012.10.008

Clark, M. K., Schoenbohm, L. M., Royden, L. H., Whipple, K. X., Burchfiel, B. C., Zhang, X., et al. (2004). Surface uplift, tectonics, and erosion of eastern Tibet from large-scale drainage patterns. *Tectonics*, 23, TC1006. https://doi.org/10.1029/2002TC001402

Cox, K. G. (1989). The role of mantle plumes in the development of continental drainage patterns. *Nature*, 342(6252), 873–877. https://doi.org/10.1038/342873a0

Davy, P., & Lague, D. (2009). Fluvial erosion/transport equation of landscape evolution models revisited. *Journal of Geophysical Research:* Solid Earth, 114, F03007. https://doi.org/10.1029/2008JF001146

Devauchelle, O., Petroff, A. P., Seybold, H. F., & Rothman, D. H. (2012). Ramification of stream networks. *Proceedings of the National Academy of Sciences of the United States of America*, 109(51), 20,832–20,836. https://doi.org/10.1073/pnas.1215218109

Dobmeier, C. J., & Raith, M. M. (2003). Crustal architecture and evolution of the Eastern Ghats Belt and adjacent regions of India. *Geological Society Special Publication*, 206(1), 145–168. https://doi.org/10.1144/GSL.SP.2003.206.01.09

Fernandes, C. C., Podos, J., & Lundberg, J. G. (2004). Amazonian ecology: Tributaries enhance the diversity of electric fishes. Science, 305(5692), 1960–1962. https://doi.org/10.1126/science.1101240

Gilchrist, A. R., & Summerfield, M. A. (1990). Differential denudation and flexural isostasy in formation of rifted-margin upwarps. *Nature*, 346(6286), 739–742. https://doi.org/10.1038/346739a0

Griffiths, D. (2018). Why does freshwater fish species richness differ between Pacific and Atlantic drainages of the Americas? Journal of Biogeography, 45(4), 784–792. https://doi.org/10.1111/jbi.13167

Karner, G. D., & Driscoll, N. W. (1999). Tectonic and stratigraphic development of the West African and eastern Brazilian margins: Insights from quantitative basin modelling. *Geological Society, London, Special Publications*, 153(1), 11–40. https://doi.org/10.1144/ GSL.SP.1999.153.01.02

Laske, G., Masters, T. G., Ma, Z., & Pasyanos, M. (2013). A 1-degree global model of Earth's crust. In Egu general assembly, 15, pp. 2658.

Leturmy, P., Lucazeau, F., & Brigaud, F. (2003). Dynamic interactions between the gulf of Guinea passive margin and the Congo River drainage basin: 1. Morphology and mass balance. *Journal of Geophysical Research*, 108(B8), 2383. https://doi.org/10.1029/2002JB001927 Lithgow-Bertelloni, C., & Silver, P. G. (1998). Dynamic topography, plate driving forces and the African superswell. *Nature*, 395(6699),

269–272. https://doi.org/10.1038/26212 Lucazeau, F., Brigaud, F., & Leturmy, P. (2003). Dynamic interactions between the Gulf of Guinea passive margin and the Congo River

- drainage basin: 2. Isostasy and uplift. *Journal of Geophysical Research*, *108*(B8), 2384. https://doi.org/10.1029/2002JB001928 Maffre, P., Ladant, J.-B., Donnadieu, Y., Sepulchre, P., & Goddéris, Y. (2018). The influence of orography on modern ocean circulation. *Climate Dynamics*, *50*(3-4), 1277–1289. https://doi.org/10.1007/s00382-017-3683-0
- McGarvey, D. J., & Ward, G. M. (2008). Scale dependence in the species-discharge relationship for fishes of the southeastern U.S.A. *Freshwater Biology*, *53*(11), 2206–2219. https://doi.org/10.1111/j.1365-2427.2008.02046.x
- Oberdorff, T., Hugueny, B., & Guegan, J.-F. (1997). Is there an influence of historical events on contemporary fish species richness in rivers? Comparisons between Western Europe and North America. *Journal of Biogeography*, 24(4), 461–467. https://doi.org/10.1111/j.1365-2699.

1997.00113.x

Osborne, L. L., & Wiley, M. J. (1992). Influence of tributary spatial position on the structure of warmwater fish communities. Canadian Journal of Fisheries and Aquatic Sciences, 49(4), 671–681. https://doi.org/10.1139/f92-076

Sacek, V., Braun, J., & Van Der Beek, P. (2012). The influence of rifting on escarpment migration on high elevation passive continental margins. *Journal of Geophysical Research*, 117, B04407. https://doi.org/10.1029/2011JB008547

Sinha, B., Blaker, A. T., Hirschi, J. J.-M., Bonham, S., Brand, M., Josey, S., et al. (2012). Mountain ranges favour vigorous Atlantic meridional overturning. *Geophysical Research Letters*, 39, L02705. https://doi.org/10.1029/2011GL050485

Straume, E. O., Gaina, C., Medvedev, S., Hochmuth, K., Gohl, K., Whittaker, J. M., et al. (2019). GlobSed: Updated total sediment thickness in the world's oceans. *Geochemistry, Geophysics, Geosystems*, 20, 1756–1772. https://doi.org/10.1029/2018GC008115

Thiéblemont, D., Callec, Y., Fernandez-Alonso, M., & Chène, F. (2018). A geological and isotopic framework of Precambrian terrains in western Central Africa: An introduction. In S. Siegesmund, M. A. S. Basei, P. Oyhantçabal, & S. Oriolo (Eds.), *Geology of Southwest Gondwana* (pp. 107–132). Cham: Springer. https://doi.org/10.1007/978-3-319-68920-3_5

Tucker, G. E., & Slingerland, R. L. (1994). Erosional dynamics, flexural isostasy, and long-lived escarpments: A numerical modeling study. *Journal of Geophysical Research*, 99(B6), 12,229–12,243. https://doi.org/10.1029/94JB00320

Turcotte, D. L., & Schubert, G. (1982). Geodynamics Applications of Continuum Physics to Geological Problems. New York: John Willey and Sons.

Acknowledgments

Jiao received support from the Natural Sciences and Engineering Research Council of Canada (Discovery Grant 2019-00243). Landscape evolution modeling was performed on the GFZ Compute-Cluster. Jiao thanks Peter van der Beek for an insightful discussion. Suggestions from Kim Huppert and Jessica Stanley and comments from Victor Sacek and anonymous reviewers made on different versions of the manuscript significantly improved the manuscript.

- Whipple, K. X., & Tucker, G. E. (1999). Dynamics of the stream-power river incision model: Implications for height limits of mountain ranges, landscape response timescales, and research needs. *Journal of Geophysical Research*, 104(B8), 17,661–17,674. https://doi.org/10. 1029/1999JB900120
- Wu, H., Kimball, J. S., Li, H., Huang, M., Leung, L. R., & Adler, R. F. (2012). A new global river network database for macroscale hydrologic modeling. *Water Resources Research*, 48, W09701. https://doi.org/10.1029/2012WR012313

Xenopoulos, M. A., & Lodge, D. M. (2006). Going with the flow : Using species discharge relationships to forecast losses in fish biodiversity. *Ecology*, 87(8), 1907–1914.

Yuan, X. P., Braun, J., Guerit, L., Rouby, D., & Cordonnier, G. (2019). A new efficient method to solve the stream power law model taking into account sediment deposition. *Journal of Geophysical Research: Earth Surface*, 124, 1346–1365. https://doi.org/10.1029/ 2018JF004867